Author's response to reviews

Title: Prevalence, Sources, and Predictors of Soy Consumption in Breast Cancer

Authors:

Carolyn A Lammersfeld (Carolyn.lammersfeld@ctca-hope.com)
Jessica King (Jessica.king@ctca-hope.com)
Sharon Walker (Sharon.walker@ctca-hope.com)
Pankaj G Vashi (pgvashi@aol.com)
James F Grutsch (jfgrutsch@yahoo.com)
Christopher G Lis (Christopher.lis@ctca-hope.com)
Digant Gupta (gupta_digant@yahoo.com)

Version: 3 Date: 9 January 2009

Author's response to reviews:

The paper is well-written overall. The study uses correct methods to analyze the data. I have one major comment on the study design.

The authors mention under the Methods section that this is a prospective consecutive case series, however, there is no prospective data presented in the paper. Did authors collect any follow-up information on soy consumption? If yes, it would be interesting to see that data. If not, the authors should revise their study design to a cross-sectional survey and not a prospective study. THANKS FOR THIS COMMENT. WE DID NOT HAVE ANY FOLLOW-UP DATA ON SOY CONSUMPTION. AS SUGGESTED, WE HAVE CHANGED THE STUDY DESIGN FROM PROSPECTIVE TO CROSS-SECTIONAL.